Author’s response to reviews

Title: Reduced flow in the left ventricle after anterior acute myocardial infarction: a case control study using 4D flow MRI

Authors:

Philip Corrado (pcorrado2@wisc.edu)
Jacob Macdonald (jamacdonald@wisc.edu)
Christopher Francois (cjfrancois@wisc.edu)
Niti Aggarwal (aggarwal.niti@yahoo.com)
Jonathan Weinsaft (jww2001@med.cornell.edu)
Oliver Wieben (owieben@wisc.edu)

Version: 2 Date: 16 Dec 2019

Author’s response to reviews:

Assistant Editor Comments:

Comment #1: Ethics Approval and Consent to Participate

-- In this section of your Declarations, please confirm whether informed consent obtained was written or verbal.

Written informed consent was obtained from the participants. This is now stated explicitly in the manuscript.

Comment #2: Authors’ Contributions

-- Currently, the contributions of authors [CJF] do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below…
We believe that CJF does meet the stated guidelines for authorship. We have revised our statement to mention his contributions that are relevant to authorship, namely his contribution to study design and his role in image scoring (analysis of data):

“CJF performed LGE image scoring and contributed to the study design.”